Author’s response to reviews

Title: Effect of propofol and etomidate on normoxic and chronically hypoxic pulmonary artery

Authors:

Nazinigouba Ouedraogo (nazinigouba@cenatrin.bf)
Bouthchi Mounkaila (m_boutchi@yahoo.fr)
Huguette Crevel (huguette.crevel@u-bordeaux2.fr)
Roger Marthan (roger.marthan@u-bordeaux2.fr)
Etienne Roux (etienne.roux@u-bordeaux2.fr)

Version: 3 Date: 21 February 2006

Author’s response to reviews:

COVER LETTER

Corresponding author:
Etienne Roux
Laboratoire de Physiologie Cellulaire Respiratoire,
INSERM E 356,
Universite Victor Segalen Bordeaux 2,
146 rue Leo-Saignat,
33076 Bordeaux cedex, France

Title of Manuscript: Effect of propofol and etomidate on normoxic and chronically hypoxic pulmonary artery

Please find attached the revised version of the manuscript MS ID#: 1076594819829246 entitled "Effect of propofol and etomidate on normoxic and chronically hypoxic pulmonary artery", with the statement that specific ethical approval of the protocole was not required by our national (French) laws. The sentence, marked in red, is added at the end of the subsection entitled “Tissue preparation”.

We hope that the revised manuscript will meet your expectations.

Dr Etienne Roux